Author's response to reviews

Title: Performance of four computer-coded verbal autopsy methods for cause of death assignment compared with physician coding on 24,000 deaths in low and middle-income countries

Authors:

Nikita Desai (nikita.desai@mail.utoronto.ca)
Lukasz Aleksandrowicz (aleksandrowl@smh.ca)
Pierre Miasnikof (MiasnikofP@smh.ca)
Ying Lu (ying.lu@nyu.edu)
Jordana Leitao (leitaojordana@googlemail.com)
Peter Byass (peter.byass@epiph.umu.se)
Stephen Tollman (Stephen.Tollman@wits.ac.za)
Paul Mee (paul.mee@wits.ac.za)
Dewan Alam (dsalam@icddrb.org)
Suresh K Rathi (RathiS@smh.ca)
Abhishek Singh (abhishek@iips.net)
Rajesh Kumar (dr.rajeshkumar@gmail.com)
Faujdar Ram (fram@iips.net)
Prabhat Jha (prabhat.jha@utoronto.ca)

Version: 3
Date: 31 October 2013

Author's response to reviews:

As per editor's request, have added competing interests, authors' contributions, acknowledgements. The numbers in the main tables have been slightly updated, though with no changes in trends or results. A track-changed version is attached.